Author's response to reviews

Title: Canadian physiotherapists’ views on certification, specialization, extended role practice, and entry-level training in rheumatology

Authors:

Linda C Li (lli@arthritisresearch.ca)
Marie D Westby (Marie.Westby@vch.ca)
Evelyn Sutton (evelyn.sutton@cdha.nshealth.ca)
Marlene Thompson (marlene.thompson@rogers.com)
Eric C Sayre (esayre@arthritisresearch.ca)
Lynn Casimiro (casimiro@uottawa.ca)

Version: 7 Date: 28 May 2009

Author's response to reviews:

Minor edits:
- Pg. 4, 2nd paragraph, 6th line: Added "To our knowledge....."
- Pg. 5, 2nd paragraph, 5th line: "Research from the UK has (instead of 'have') shown....."